Author's response to reviews

Title: Effects of Multiparity and Prolonged Breast-Feeding on Maternal Bone Mineral Density: A Community-Based Cross-Sectional Study

Authors:

Janaka Lenora (Robolge.Lenora@med.lu.se)
Sarath Lekamwasam (sarithlk@slt.net.lk)
Magnus K Karlsson (Magnus.Karlsson@med.lu.se)

Version: 3 Date: 24 June 2009

Author's response to reviews: see over
June 24, 2009

The Editor-in-chief, BMC Women’s Health,

Dear Sir,

Manuscript ID: 2526767724428534
Effects of Multiparity and Prolonged Breast-Feeding on Maternal Bone Mineral Density: A Community-Based Cross-Sectional Study

We thank you for accepting our paper for publication in your journal. We have gone through the manuscript and corrected the suggested format changes as follows,

Title: - We have corrected as “Effects of Multiparity and Prolonged Breast-Feeding on Maternal Bone Mineral Density: A Community-Based Cross-Sectional Study”

Author list: - We have changes the names of authors as it is in the submission system. Order of the author affiliations also reorganized.
Table Size: - Now we have arranged tables to fit into an A4 portrait page.

Subheading in Title page: - We have removed.

We have also checked the manuscript for typographical errors.

Yours sincerely,

Janaka Lenora  
Sarathe Lekamwasam  
Magnus K Karlsson

Corresponding Author: -  
Janaka Lenora  
Clinical and Molecular Osteoporosis Research Unit,  
Department of Orthopaedics,  
Malmö University Hospital, Lund University,  
SE-20502, Malmö, Sweden  
Phone- +46-40 337517  Fax - +46-40336200

E mail. Robolge.Lenora@med.lu.se